Author's response to reviews

Title: Deep Brain Stimulation for Treatment-Refractory Obsessive Compulsive Disorder: a Systematic Review

Authors:

Sina M Kohl (sina.kohl@uk-koeln.de)
Deva M Schoenherr (devamari@gmx.net)
Judy Luigjes (j.luigjes@amc.uva.nl)
Damiaan Denys (ddenys@gmail.com)
Ulf Mueller (ulf.mueller@med.ovgu.de)
Doris Lenartz (doris.lenartz@uk-koeln.de)
Veerle Visser-Vandewalle (veerle.visser-vandewalle@uk-koeln.de)
Jens Kuhn (jens.kuhn@uk-koeln.de)

Version: 2
Date: 27 March 2014

Author's response to reviews:

Dear Editor,

we included the line numbering and added an ethical statement under methods.